Author's response to reviews

Title: Factors associated with involuntary admissions among non-psychotic patients with substance use disorders and comorbidity: a cross-sectional study

Authors:

Anne Opsal (anne.opsal@sshf.no)
Oistein Kristensen (Oistein.Kristensen@sshf.no)
Tor K Larsen (tkmaclarsen@me.com)
Gro Syversen (syversen.gro@gmail.com)
Elise BA Rudshaug (eliseaasen@hotmail.com)
Arne Gerdner (Arne.Gerdner@hhj.hj.se)
Thomas Clausen (thomas.clausen@medisin.uio.no)

Version: 4 Date: 15 January 2013

Author's response to reviews: see over
Dear, Ms Flory Mae Calumpita
on behalf of Dr Roberto Forero,

Manuscript MS: 1959391252775934 - Factors associated with involuntary admissions among non-psychotic patients with substance use disorders and comorbidity: a cross-sectional study

Thank you for your support and interest in our paper and for allowing us to resubmit a revised version for publication in BMC Health Services Research.

We appreciate the valuable and constructive comments from the editor and have revised the paper according to your suggestions.

We hope the revised paper is now acceptable. Please do contact me again if additional issues need to be addressed.

Detailed responses are provided below.

Yours Sincerely,

Anne Opsal
(on behalf of the authors)
Sørlandet Hospital, Addiction Unit
Kristiansand, Norway
Associate Editor's comments:

On Page 15, conclusions section, second paragraph:

Replace "Female gender, receiving public welfare benefits, frequent visits to physicians for somatic complaints, and drug injection during the past 6 months were all associated with involuntarily admission pursuant to the Social Services Act" with "Women, receiving public welfare benefits, with frequent visits to physicians, and drug injection during the past 6 months were more likely to be associated with involuntarily admissions to health services in accordance with the Norwegian Social Services Act of 1993."

We have replaced the text according to Associate Editor’s comments.